Author's response to reviews

Title: Pay clinicians to join clinical trials? A review of guidelines and survey of trialists

Authors:

James Raftery (raftery@soton.ac.uk)
Christine Kerr (d.kerr@btinternet.com)
Sheila Hawker (sheilahawker@gmail.com)
John Powell (john.powell@warwick.ac.uk)

Version: 2 Date: 10 June 2008

Author's response to reviews:

Hi
I have revised the abstract, included a reference to Ethics Committee approval for the survey and reformatted the references, as instructed by your email. I hope this is sufficient.
Best wishes
James Raftery